Author's response to reviews

Title: Use of Dried Blood for Measurement of Trans Fatty Acids

Authors:

Ruby Gupta (g_ruby2123@yahoo.co.in)
Ransi Ann Abraham (ransijob@gmail.com)
Savita Dhatwalia (cardiacbiochem@gmail.com)
Lakshmy Ramakrishnan (lakshmy_ram@yahoo.com)
Dorairaj Prabhakaran (dprabhakaran@ccdcindia.org)
Kolli Srinath Reddy (ksreddy@ccdcindia.org)

Version: 2 Date: 16 March 2009

Author's response to reviews:

Dear Editor,

We have revised the 'abstract' as well as 'manuscript' as per formatting requirement of your journal and resubmitting the same. The manuscript is accompanied with one table and two figures. Please consider.

thanking you,

Dr. R. Lakshmy